Author’s response to reviews

Title: Implementation of Gestational Weight Gain Guidelines - what's more effective for ensuring weight recording in pregnancy?

Authors:
Shelley Wilkinson (shelley.wilkinson@mater.org.au)
Michael Beckmann (michael.beckmann@mater.org.au)
Elin Donaldson (elin.donaldson@mater.org.au)
Sally McCray (sally.mccray@mater.org.au)

Version: 2 Date: 10 Dec 2018

Author’s response to reviews:

Thank you for accepting our manuscript for publication conditional on the minor additional edits.

We have proof read the paper and have also corrected the sentence, as per Reviewer 1's comments.

We have also updated the ethical statement to read: Consent was not required from patients. This project methodology was assessed to meet the requirements of the National Statement on Ethical Conduct in Human Research (2007) and ethical approval was obtained from the hospital’s Human Research Ethics Committee (HREC/14/MHS/119. Amendment AM01).